Author’s response to reviews

Title: Factors Influencing Decision Making in Orthopedic Surgery - an international online survey of 1147 orthopedic surgeons

Authors:

Anders Jönsson (anders.k.jonsson@vgregion.se)
Richard Kasch (kaschr@uni-greifswald.de)
Jettoo Prithee Prithee (prithee.jettoo@nhs.net)
Mohit Bhandari (bhandam@mcmaster.ca)
Arndt P Schulz (schulz@biomechatronics.de)

Version: 3 Date: 20 March 2012

Author’s response to reviews:

1. Abstract: corrected

2. Ethics: Added

3. Competing interests: Added

4. Authors' information: provided

5. Acknowledgement: added